Author's response to reviews

Title: A Systematic Review of Suicide Prevention Interventions targeting Indigenous peoples in Australia, United States, Canada and New Zealand.

Authors:

Anton Clifford (Anton.Clifford@iuih.org.au)
Christopher Doran (Christopher.Doran@hnehealth.nsw.gov.au)
Komla Tsey (komla.tsey@jcu.edu.au)

Version: 2 Date: 25 October 2012

Author's response to reviews:

25 October 2012

The Editorial Team
BMC Public Health

Dear Editorial Team,

As per your request, the manuscript has been revised confirming that it is a systematic review (as per revised title) and adheres to the PRISMA guidelines (page 6, paragraph 2 of manuscript).

In addition, please note that Figure 1 is a flowchart of the systematic search strategy, consistent with the requirements of a systematic review.

Regards,

Anton Clifford